



Chaopeng Guo 🔍, Yiming Liu, Meiyu Na and Jie Song \* 🕑

Software College, Northeastern University, Shenyang 110000, China; guochaopeng@swc.neu.edu.cn

\* Correspondence: songjie@mail.neu.edu.cn

Abstract: Blockchain techniques have been introduced to achieve decentralized and transparent traceability systems, which are critical components of food supply chains. Academia and industry have tried to enhance the efficiency of blockchain-based food supply chain traceability queries. However, the cost of traceability queries remains high. In this paper, we propose a dual-layer index structure for optimizing traceability queries in blockchain, which consists of an external and an internal index. The dual-layer index structure accelerates the external block jump and internal transaction search while preserving the original characteristics of the blockchain. We establish an experimental environment by modeling the blockchain storage module for extensive simulation experiments. The results show that although the dual-layer index structure introduces a little extra storage and construction time, it significantly improves the efficiency of traceability queries. Specifically, the dual-layer index improves the traceability query rate by seven to eight times compared with that of the original blockchain.

Keywords: traceability system; food supply chain; traceability query; blockchain; indexing



Citation: Guo, C.; Liu, Y.; Na, M.; Song, J. Dual-Layer Index for Efficient Traceability Query of Food Supply Chain Based on Blockchain. *Foods* **2023**, *12*, 2267. https://doi.org/ 10.3390/foods12112267

Academic Editors: Guendalina Graffigna, Luigi Lucini and Antonello Santini

Received: 17 April 2023 Revised: 25 May 2023 Accepted: 2 June 2023 Published: 5 June 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

# 1. Introduction

Ensuring food safety is crucial to maintaining people's quality of life. The food supply chain consists of the entire process of food production: processing, distribution, consumption, and disposal [1,2]. However, the lack of transparency and traceability can lead to large-scale food fraud affecting millions of people [3]. Transparency can enhance consumers' confidence in food quality and safety, and traceability can be used to verify compliance with the food circulation process and prevent contamination or fraud in the supply chain [4]. Therefore, it is essential to ensure the transparency and traceability of the historical data in the food supply chain.

Traceability enables stakeholders to track products from the farm to the fork, ensuring safety and high quality [5]. Traceability queries for historical data can trace products in the supply chain and track the circulation of products [6]. For example, in the food supply chain, consumers can use traceability queries to obtain the source of food-related issues and any subsequent health problems they may have experienced. By providing a clear view of the entire supply chain, traceability helps to quickly identify issues, isolate contaminated products, and prevent the further spread of contamination or fraud.

Traditional centralized traceability systems have several issues, such as data monopoly, low security, untrustworthiness, low efficiency of data sharing, lack of transparency, and difficulty in meeting the needs of modern society [7]. Blockchain provides a new solution for traceability systems. The distributed storage structure of blockchain ensures the authenticity and integrity of information. Moreover, the decentralized platform and consensus mechanism prevent malicious tampering with information collection and circulation [8].

Blockchain can be broadly defined as a distributed database that shares and maintains data consistency. It consists of a connected sequence of blocks that store data in a block-and-chain structure. Transactions with timestamps are protected by public key encryption and verified by the community. There are four key features of blockchain. Firstly, it is decentralized. Transactions can be conducted between any two clients in the network without a central server or a third-party agent. Secondly, blockchain is persistent. Transactions cannot be altered after being verified and recorded [9]. Thirdly, blockchain is anonymous. Users can generate one or multiple anonymous addresses to conduct transactions, effectively avoiding personal identity exposure. Finally, blockchain is auditable. With every transaction being authenticated and recorded with a timestamp, users can track and verify records from any node in the network, enhancing the traceability and transparency of the stored data [10]. For a more detailed introduction to blockchain technology and its applications in the supply chain, please refer to the review article by Sunny et al. [11].

Blockchain provides traceability capabilities [12]. In the storage structure of a blockchain, the block header data contain the hash value of the previous block, which can be used to retrieve the previous block and enable retrospective query functionality through traversal. However, blockchain is not optimized for traceability or conditional queries, resulting in extremely low query efficiency [13]. To query records, blockchain has to traverse all records to generate the final query results [14]. Many blockchain systems, such as Fabric, Ethereum, and Parity, distinguish the current and historical state of data. Data related to various events are written in the form of key–value pairs. For key k, the data include the latest pair of the current state of k and all pairs of the historical state of k. The collection of current states for all keys is denoted as state-db, while the collection of the historical states is denoted as history-db. The history-db data are time-dependent in nature, and analysis of these data can generate valuable business insights and support various use cases [10]. In this study, we treated the query method of obtaining these historical data as traceability queries. Notably, traceability queries in the native blockchain must be conducted by traversing blocks, which is time-consuming.

Current research has mainly focused on blockchain conditional query optimizations, involving two research directions. The first is enriching the query function and improving the query efficiency with external databases. However, these approaches lose blockchain security and require additional storage management and cost. The second one is altering the block structure to achieve lower storage space overhead and keep the security mechanism. Generally, a blockchain traceability query includes two phrases: an internal block query and an external block jump. The internal block data query searches the data according to given query conditions, for example, the produce ID within a block. The external block jump searches the previous block that contains the corresponding historical records. Improvements in the conditional query can optimize the internal block query performance but not the external block jump. Therefore, the current blockchain conditional query optimization methods comprehensively cannot boost traceability queries.

In this study, we followed the structure alternation direction to optimize the traceability query performance by introducing a dual-layer index structure. The dual-layer index structure includes an internal index and an external index. The internal index reconstructs the storage structure of the block by combining a Merkle tree and a B+ tree to accelerate the internal block data query. The external index alters the storage structure of the transaction. It contains a temporary table of traceability objects to realize a direct jump between blocks according to the traceability target to optimize the external block jump. Through the combination of the internal index and the external index, the dual-layer index structure not only optimizes the conditional query of blockchain but also enhances traceability queries without losing the characteristics of blockchain.

The main contributions of this study are as follows:

(1) We addressed the issue of low traceability query efficiency in blockchain-based traceability systems and avoided the tamper resistance loss caused by introducing external databases. Our solution ensures both efficient querying and the integrity of the blockchain.

(2) The dual-layer index accelerates the two steps of traceability data retrieval, namely internal block search and external block jump. It comprehensively and effectively improves traceability query efficiency.

(3) The experimental results demonstrate that the dual-layer index structure improves traceability query efficiency by approximately seven to eight times compared with that of the native Merkle tree, making it suitable for blockchain-based traceability systems.

(4) We abstracted the storage layer of blockchains to build a blockchain storage module, providing a tool for stimulation experiments, in which we identified the jump cost between blocks in the traceability process as a factor affecting traceability efficiency, providing ideas for the subsequent optimization of traceability efficiency.

The organization of the paper is as follows: Section 2 summarizes related studies, including blockchain-based food supply chain studies, general supply chain studies, and blockchain-related query efficiency optimization approaches. Section 3 presents the methodology in our research. Section 4 describes the design of the dual-layer index structure in detail, and Section 5 illustrates the operations of the dual-layer index. Section 6 shows the experimental setup and results. Section 7 presents the discussion of our findings, including experimental result analysis, a comparison with related methods, implementation details, and a summary of the assumptions and limitations of our method. Section 8 concludes the paper.

## 2. Related Studies

The traditional centralized traceability system in the food supply chain lacks transparency and traceability. In contrast, blockchain technology has the potential to meet the transparency and traceability requirements of the food supply chain. However, the traceability query efficiency in blockchain can be a limiting factor for the overall efficiency of the food supply chain. This section introduces existing supply chain optimization methods based on blockchain and demonstrates the feasibility of using blockchain technology in the food supply chain. Furthermore, this section discusses optimization methods that rely on external databases and structural changes and compares their advantages and differences with those of the approach proposed in this paper.

#### 2.1. Food Supply Chain Based on Blockchain

Prashar et al. [15] proposed a blockchain-based solution that eliminates the need for a secure centralized structure, resulting in a transparent, accurate, and traceable supply chain system. Damoska Sekuloska et al. [16] developed a three-layer model for a smart local food supply chain that enhances food accessibility, traceability, and safety. Zhang et al. [17] developed a trusted traceability system for grain and oil food using the Hyperledger Fabric framework. The system addresses issues related to low data security and poor data sharing while also improving traceability for producers by leveraging Ethereum to provide a secure, transparent, and efficient solution for tracking and tracing products in the supply chain.

The above studies have confirmed that blockchain technology provides a feasible technical solution for the food supply chain, which can improve the traceability, transparency, and safety of the food supply chain. However, the goals of the above studies slightly differ from ours. In this study, we tried to optimize the performance of traceability queries in the blockchain-based food supply chain while keeping the blockchain features.

### 2.2. Other Supply Chain Based on Blockchain

Kong et al. [18] proposed BCSChain based on blockchain for the ceramic supply chain system. BCSChain aims to address the issues of transparency and security prevalent in traditional ceramic supply chains. Hader et al. [19] introduced a novel traceability framework for the textile supply chain based on blockchain technology. This framework allows a transparent information-sharing platform among all supply chain members. Abdallah et al. [20] proposed a blockchain-based framework for the online sale of pharmacutical products. The system keeps track and informed about sale transactions and ensures secure payment dispersal and medicine delivery. Liu et al. [21] developed a financial management platform that merges blockchain and supply chain technologies. By utilizing blockchain credit transfer, information sharing, data traceability, and tamper-proof features, this platform aims to solve critical issues such as the inability to transfer core enterprise credit.

Several researchers have proposed new solutions that leverage the unique characteristics of blockchain. The results of these studies have demonstrated that blockchain can effectively address the security and traceability issues inherent in traditional supply chain systems.

#### 2.3. Optimization Methods Based on External Database

One direction is to integrate database techniques into blockchain. Helmer et al. [22] proposed EternityDB, integrating database functions into blockchains via intelligent contracts, in which databases and data are moved to the blockchain. However, copying and storing such large databases on each node of a peer-to-peer network not only wastes storage space but also increases storage pressure and creates cost problems. BigChainDB [23] combines document-based NoSQL database functions to achieve the fast query and reliability of blockchain but lacks SQL support. Muzammal et al. [24] proposed ChainSQL, which supports SQL operations. However, like BigChainDB, their actual data reside in an off-chain database. Namely, users must trust their connected peers and ensure the database stores accurate data because the query results might not be consistent with the blockchain.

Another direction is to embed middleware in the blockchain to support database operations. Zhou et al. [25] proposed Ledgerdata to improve the query efficiency of super ledger blocks and transactions by adding data analysis middleware to Hyperledger Fabric. However, its application scope is limited to Fabric. Li et al. [26] proposed EtherQL, a query layer on the blockchain, which provides an efficient query primitive to expand query function and improve query efficiency. Fa et al. [27] further expanded and enhanced the query expansion function. However, this approach requires that the server always returns correct results, which is too restrictive in real-world scenarios. Peng et al. [14] proposed a Verifiable Query Layer (VQL) that adopts a three-tier architecture, including the underlying blockchain system, middleware layer, and application layer. The middleware provides efficient and verifiable query services to ensure the authenticity of the query results of the blockchain. However, the MongoDB introduced by the middleware layer still needs additional management, leading to storage overhead.

Optimizing the query efficiency of blockchains using external databases can lead to the loss of the security characteristics of the blockchain. Additionally, it introduces storage management costs and reduces transparency in the food supply chain, which are unacceptable. As a result, some researchers have proposed optimization strategies based on structural alteration to address these issues. Namely, these strategies improve query efficiency while maintaining the security and transparency characteristics of the blockchain.

#### 2.4. Optimization Methods Based on Structure Alternation and Novel Data Forms

Introducing external query structures to alter the blockchain structure is an essential method for improving the performance of blockchain queries. Ma et al. [28] designed a blockchain-based log system with lightweight, compatible modules for existing blockchain platforms. This system employs a hierarchical timestamp structure that enables efficient range queries on timestamp fields. Ren et al. [29] proposed a Bitcoin-based model for Internet of Things data queries that uses the dual-combination Bloom filter (DCOMB) query method. This method transforms the hash computing capacity used for Bitcoin mining into query computing power, enhancing transaction query proficiency.

Another direction for optimizing the efficiency of blockchain is to improve the storage structure of blocks. Pei et al. [30] introduced an indexing technique based on Merkle Semantic Trie. They extracted semantic information from the data in the chain to construct an indexed structure on the consensus chain, allowing for the real-time and effective query of data on and off the chain. Niu et al. [31] improved the efficiency of transaction search by reorganizing the blockchain and constructing an index structure using prefix tree and

ODS technology. This scheme enables efficient, privacy-preserving transaction queries based on the transaction hash and address, even for lightweight clients. He et al. [32] proposed the T-Merkle tree, a novel blockchain storage structure that improves the design of the Merkle tree, reduces storage costs, and enhances query efficiency. Xu et al. [33] proposed the MPT-Chain tree, combining the characteristics of the Patricia tree and Merkle tree. MPT-Chain guarantees distributed index consistency and query result accuracy.

The inherent characteristics of blockchain, including transparency and traceability, are crucial to food supply chains. Our research idea is based on optimizing the traceability query of historical data by modifying the storage structure of the blockchain while retaining its fundamental characteristics. To achieve this, we reconstructed the storage structure of block data and added both internal and external indices.

In contrast to previous methods [30–33], our approach not only enhances query efficiency by adding indices within each block but also complements this with external indices. We further leveraged index fields in a temporary traceability table and leaf nodes within blocks to facilitate jumping to the destination block and avoid sequential queries, further improving the traceability query speed.

The other optimization direction is to utilize novel data forms. New and emerging data formats can potentially improve the analysis process [34]. The storage format of products in the supply chain can be simplified as spatiotemporal data, where temporal data record the time of product changes, and spatial data record the flow direction of products. With spatiotemporal data, the system can support context queries, and anomalies in supply can be easily found. Differently, in this study, we mainly focused on traceability performance rather than context queries.

In summary, our approach provides an efficient solution for blockchain-based traceability systems that ensures both querying efficiency and the integrity of the blockchain.

#### 3. Methodology

In this section, we describe our research methodology, including our motivation, chosen approach, idea overview, and experimental design.

### 3.1. Motivation

Blockchain technology is known for certain key characteristics, including distribution, decentralization, security, nontampering ability, and traceability, that can effectively address the trust issues experienced with traditional traceability systems. However, these benefits can be limited by the inefficient querying capability of the blockchain system. Therefore, our research objective was to develop an optimization approach to enhance the efficiency of blockchain traceability queries.

## 3.2. Chosen Approach

Previous research can be categorized into two directions: incorporating an external database for efficient querying through database technology and introducing an index structure to enhance efficiency by modifying the block structure. However, using an external database raises security concerns and undermines the security features of blockchain. This violates the security requirements of supply chain traceability systems. Instead, we propose introducing an index structure and modifying the blockchain structure to enhance query efficiency while maintaining the fundamental characteristics of blockchain.

#### 3.3. Idea Overview

The data traceability process in blockchain involves two steps: traversing all blocks to find the target block and then searching all data within that block. To optimize these steps, we propose a dual-layer index structure that limits the number of traversed blocks and the amount of data traversed within blocks. The dual-layer index structure utilizes an external index to facilitate external block jumps and an internal index to enable internal block query.

### 3.4. Experimental Design

We used a quantitative analysis method to evaluate the effectiveness of our proposed dual-layer index structure in improving query efficiency through experimental validation. Because using index structures may introduce time and space overhead, we designed comparison experiments to evaluate the performance of the dual-layer index structure against that of the native block structure in terms of time overhead, space overhead, and query efficiency. To account for variations in data amount and block size, we created an experimental dataset with varying values of these parameters. Similarly, we create another experimental dataset that varies the block depth to investigate its impact on query efficiency. In addition, we constructed a large-scale dataset and multiple complex scenarios to validate the performance of the dual-layer index under more complicated condition scenarios. Based on the experimental results, we analyzed the impact of the data amount and block size on the time and space overhead of the dual-layer index structure as well as the influence of data amount, block size, and block depth on query efficiency. We evaluated the performance improvement of the traceability query using the dual-layer index structure from these results and identified its applicable scenarios and limitations.

### 4. Dual-Layer Index Structure

The block storage structure of blockchains is designed to ensure data accuracy through Merkle trees or their variants [35], because any modification to the block structure alters the header hash value, causing a mismatch with the next block. Moreover, transaction existence is verified by expanding the "light nodes" in a public blockchain environment [36]. Typically, the Merkle tree is integrated into the transaction data of a block, which consists of two main parts: block header and block body. The block header consists of the hash value of the previous block (*preHash*), the Merkle tree root, and the hash value of the current block (*curHash*). The block body contains the Merkle branch segment. Traceability queries can be performed by recursively searching for the previous block, starting from the last block that contains the newest traceability transaction using *preHash*. However, given the high search cost associated with scanning all blocks, this approach severely limits the performance of the traceability query system.

To support fast traceability queries, a dual-layer index structure is proposed. The dual-layer index structure consists of an external index and an internal index. The external index allows for direct jumps between traceability-related blocks, while the internal index enables fast transaction data queries inside a block. Utilizing both the external and internal index, the dual-layer index structure can significantly improve the query efficiency of blockchain-based traceability systems.

## 4.1. External Index Design

Using the food supply chain as an example, each product is assigned a unique identifier, such as a string of bar codes, QR codes, or other forms stored within the blockchain. Additionally, each transaction within the blockchain is assigned a unique TransactionID, which can be used to differentiate between transactions. In a native blockchain, traceability query information is accessed by comparing the TransactionID and cross-referencing product identifiers to identify the desired information. In the dual-layer index structure, a TraceID is used to query the blockchain. The TraceID is the storage form of the product identifier mentioned earlier. The term "TraceID" is used interchangeably with "product identifier" to improve clarity.

He et al. [32] proposed a T-Merkle tree and showed that intermediate nodes could store the necessary data by modifying the node storage structure. Based on this idea, we achieve part of the external index in the block body. The external index consists of two parts: an item inside the blocks of the blockchain and a temporary traceability object table. The leaf nodes of a block with the external index contain more comprehensive data than the native structure, including TransactionID, TraceID, Time, *preTraceHash*, Transaction, and Hash. *preTraceHash* refers to the hash value of that specific block, where the previous

transaction data with the same TraceID are stored. By means of *preTraceHash*, we can directly jump to the previous block location containing the transactions with the same TraceID. A temporary traceability object table is introduced to find the newest traceability block when we update a block or conduct a traceability query. By changing the storage structure in the blockchain, sequential retrieval in the traceability query becomes a direct jump to the traceability-related blocks.

The external index overview is shown in Figure 1. Figure 1a shows the Merkle tree with the external index item. The leaf node (T1) stores TransactionID, TraceID, Time, *preTraceHash*, Transaction, and Hash. Both the intermediate node (H12) and root node (H1234) compute and store their own hash from the hash of their child nodes. Figure 1b shows a temporary table that stores the TraceID, TransactionID, and *preTraceHash*. By maintaining this temporary traceability object table, the initial traceability block can be quickly located.



(b) temporary traceability object table

Figure 1. External index overview.

Compared with the native Merkle tree, a Merkle tree with the external index structure shown in Figure 1 directly stores hash values at the leaf nodes, resulting in a shallower tree. This approach can accelerate the process of block jump during a traceability query by including the block hash value *preTraceHash* in the leaf node and maintaining the temporary traceability object table of traceability objects. However, implementing this external index structure for different blockchain types (e.g., Bitcoin, Ethereum, and Hyperledger Fabric) requires additional steps, such as adding temporary traceability tables and fields to transaction data.

In summary, the implementation of the external block index involves two parts: altering the storage structure of block internal transaction data and adding *preTraceHash* in the leaf node to enable a direct jump, and creating a temporary table to facilitate the fast retrieval of the newest traceability blocks to enhance the traceability query speed.

## 4.2. Internal Index Design

The B+ tree was shown to improve the speed of sequential lookup and support fast queries while reducing disk I/O [37]. Given the advantages of both the B+ tree and Merkle tree, we designed an MB+ tree (Merkle and B+ Tree) to support efficient traceability queries in the block. The MB+ tree leverages the tamper-proof features of Merkle trees to enable fast verification of blocks while also taking advantage of the efficient query and location capabilities of B+ trees to perform efficient searches of transaction data within blocks.

As shown in Figure 2a, the MB+ tree has all the characteristics of the B+ tree and the Merkle tree.

(1) In the B+ tree, keywords connect non-leaf nodes to their corresponding child nodes. Similarly, in the MB+ tree, TraceID is used as the keyword. (2) In the Merkle tree, leaf nodes store data, while other nodes store the hash of their child nodes. However, in the MB+ tree, the leaf node stores additional information, such as the TraceID, namely, the keyword required by the B+ tree, the *preTraceHash* required by the external index, and the hash value calculated by all the leaf node fields. The nonleaf nodes in the MB+ tree store pointers to their child nodes, keyword ranges, and hash values calculated by hashing all child nodes.



(b) The Leafnode of MB+ Tree in Bitcoin

(c) The Leafnode of MB+ Tree in Fabric

Figure 2. Block structure with internal index.

Specifically, in the MB+ tree, there are two types of nodes: leaf nodes and intermediate nodes (including the root node). Leaf nodes in the MB+ tree are responsible for storing all data. An example of the MB+ leaf node in Bitcoin is shown in Figure 2b, which consists of TransactionID, TraceID, Time, *preTraceHash*, Transaction, and Hash. The intermediate and root nodes in the MB+ tree (Figure 2a) are responsible for storing keyword information (2, 4, 6, 7, and 10), pointers (p1, p2, and p3), and hash values (H1–6) composed of keyword information, pointers, and child nodes.

In summary, based on the Merkle tree, we used the characteristics of the B+ tree to construct the MB+ tree to build the internal index. The keyword information and pointers are stored in the root and intermediate nodes. The leaf nodes store the block hash value (*preTraceHash*) of the previous traceability information and the fields of the TraceID as external index structures. By utilizing the pointers and keyword information, the traceability transaction data can be quickly located within a block.

Due to the different internal storage structures of specific blockchain architectures, such as Bitcoin, Ethereum, and Hyperledger Fabric, the internal indices of the three blockchains are discussed below. In Bitcoin, the MB+ tree is utilized as shown in Figure 2a. The native Bitcoin leaf node stores transaction data, while the intermediate nodes, including the root node, store hash values. The MB+ tree leaf node, shown in Figure 2b, contains the TransactionID, TraceID, Time, Transaction, *preTraceHash*, and Hash. Additionally, the intermediate node stores a pointer that can point to the subnode and the keyword information representing the storage range of the su-node. Moreover, the leaf nodes are sequentially arranged and connected with a chain pointer. This design enables sequential data retrieval in the block based on the order of the TraceID.

The difference between using the MB+ tree in Ethereum and Bitcoin lies only in some of the data categories at the head of the block. However, the final MB+ tree data structure is the same as that for Bitcoin.

In Hyperledger Fabric, the FB+ tree is utilized as shown in Figure 2c. Similar to Bitcoin, the leaf node stores transaction data with additional details of signatures and transaction proposals. However, in contrast with Bitcoin, it no longer stores hash values. Figure 2c demonstrates the transaction content with additional details regarding the signature and transaction proposal. The intermediate and root nodes are responsible only for storing keyword information and pointers, with the keyword information represented by TraceID.

In summary, the implementation of the internal block index is primarily based on the features of the B+ tree, which was enhanced to form the MB+ tree. In the MB+ tree, the root and intermediate nodes store keyword information and pointers. The leaf nodes store the block hash value of the block where the previous traceability object is located along with an external index structure containing TraceID. Using pointers, the MB+ tree enables rapid retrieval of traceability transactions from a block.

#### 5. Dual-Layer Index Operation

To take advantage of the dual-layer index structure, we altered the storage structure in Bitcoin, Ethereum, and Hyperlegder Fabric. Both Bitcoin and Ethereum use MB+ trees to store transaction data. Meanwhile, the FB+ tree is used in Hyperledger Fabric. The critical difference between the FB+ and MB+ trees is that the former no longer stores the hash value and omits the process of calculating the hash value, while most other structures remain unchanged. All three improved blockchains share the same external index structure. Therefore, creating and querying these indices are similar. In this section, we only consider Bitcoin as an example to illustrate the operations of the dual-layer index.

#### 5.1. Dual-Layer Index Creation

During index creation, if the amount of data does not reach a predefined threshold, the data are initially stored in a cache table. The threshold is the maximum amount of transaction data a block can contain. Once we have enough initial data, a new block is created and linked to the previous block in the blockchain. The creation of the dual-layer index structure involves the creation of internal and external indices. In the following sections, we describe these two creation processes in detail.

#### 5.1.1. Internal Index Creation

The creation of the internal index involved creating an MB+ tree, in which nodes are arranged in an order set *M*. Building an MB+ tree includes three steps:

(1) When the first transaction is inserted into a new block, the system queries the temporary traceability object table associated with the external index structure of the block to obtain *preTraceHash*. Next, the system embeds the transaction data and *preTraceHash* into a leaf node, which becomes a root node later in the new block. During this process, the leaf node hash value *Leaf NodeHash* is calculated based on the TransactionID, TraceID, Time, *preTraceHash*, and Transaction.

(2) When the other transaction data are inserted, the system queries the temporary traceability object table associated with the external index structure to obtain *preTraceHash*. The system embeds the transaction data into a leaf node and determines their insertion

position based on the TraceID. If the number of keywords in the current node is less than or equal to |M| - 1, the insertion is completed. Otherwise, the leaf node must be split into left and right leaf nodes. The left leaf node contains the data stored in the first |M|/2 leaf nodes, and the right leaf node stores the remaining data. The middle keyword, located at position |M|/2 + 1, is assigned to the parent node's keyword. The parent node's left pointer points to the left leaf node, and its right pointer points to the right leaf node.

(3) For the intermediate node, if the number of keywords is less than or equal to |M| - 1, then the insertion is complete. Otherwise, the system must split the middle node into two intermediate nodes. The left intermediate node contains the first |M - 1|/2 keywords, and the right node contains the remaining |M| - (|M - 1|/2) keywords. The middle keyword, located at position |M|/2, is assigned to the parent node's keyword. The parent node's left pointer points to the left intermediate node, and the right pointer points to the right intermediate node. The current node pointer is set to the parent node, and the system returns to Step (2) to insert the next transaction until all transactions are inserted. For hash values stored in intermediate and root nodes, the system uses the Pointers, ChildNodeHashs, and TraceID to calculate the hash value. The system uses the blockchain version number, block index number, timestamp, difficulty calculation, the hash value of the previous block, and MB+ tree root to calculate the current block hash.

#### 5.1.2. External Index Creation

Each transaction inserted in the block must have a TraceID attribute. When we create the external index, the temporary traceability object table is first queried. If TraceID does not exist, then the TraceID is inserted into the temporary traceability object table. If the TraceID exists, we must update *preTraceHash* in the table pointing to the block with the newest traceability transaction.

To facilitate queries, we store the temporary traceability object table in a B+ tree. The insertion operation is similar to establishing an MB+ tree inside the block. Therefore, we do not further elaborate on the insertion algorithm's operation.

## 5.2. Dual-Layer Index Query

The query process of the dual-layer index structure is divided into two parts, internal and external index query. The internal index queries the data in blocks, and the external index queries the temporary traceability object table. The overall traceability query process consists of two steps:

(1) We first query the temporary traceability object table according to the TraceID to find the latest transaction location. If the TraceID does not exist, an empty result is returned. Otherwise, the block with *preTraceHash* is searched.

(2) While searching the MB+ tree of the block, we try to find a leaf node with the same TraceID. Firstly, we determine whether the current node is a leaf node. If it is not a leaf node, the search is continued in the subtrees according to the relationship between the TraceID and the keyword of the current node. Then, if a leaf node is found, *preTraceHash* is evaluated. If *preTraceHash* is empty, the transaction in the leaf node is stored in the result set, which is returned to the client. Otherwise, Step (2) is executed according to *preTraceHash*.

#### 6. Experiments

In this experiment, we compared the original non-layer index and the dual-layer index structure in terms of construction cost and traceability query efficiency. In this section, we introduce the software and hardware environment used during the experiment. Then, the experimental design and experimental results are presented.

## 6.1. Environment Setup

The native blockchain involves various complex components, such as consensus and cryptography. We want to clarify that these topics were beyond the scope of this study. Therefore, we built a simulation platform to conduct our experiment for the following reasons:

(1) The dual-layer index aims to alter the blockchain storage layer, not the entire blockchain framework. If the original blockchain framework was used, the cost of the whole system must be considered, including node consensus, block synchronization, etc. This would introduce more factors that could influence the analysis of experimental results.

(2) There is currently no available block storage module that solely focuses on blockchain storage, highlighting the need for this research.

#### 6.2. Dataset Setup

The aim of the experiment was to evaluate the effectiveness of the dual-layer index structure and explore the factors that impact its construction time, storage, and query performance. Because these factors are related to the blockchain structure and not to the specific features of blockchain data, we generated the dataset used in the experiments. To simulate block data, the data format was designed to match that of the Bitcoin format, including a block header and body. The internal fields of the block header are consistent with Bitcoin's, and the block's internal structure is based on the MB+ tree. The specific attribute values of the transaction stored in the block did not affect the study purpose and were set as fixed-length random strings.

The amount of data is the most intuitive factor that affects query efficiency. We considered the number of traceability transactions, namely, the data amount, as the influencing factor and recorded it as  $d_i$ . The datasets used for our experiment are shown in Table 1.

Table 1. Configuration of data amount.

$d_1$	$d_2$	<i>d</i> <sub>3</sub>	$d_4$	$d_5$	$d_6$	$d_7$	<i>d</i> <sub>8</sub>
$10^{4}$	$10^{5}$	$2 imes 10^5$	$5 imes 10^5$	106	$2 imes 10^6$	$5 imes 10^6$	107

Traceability queries consist of two steps, namely, data query within blocks and jumping between blocks. As such, the amount of traceability transaction data within blocks also affects the efficiency of the query. Therefore, the amount of traceability data within blocks, namely, the block size, is another influencing factor and is denoted as  $b_j$ . The block size is shown in Table 2.

Table 2. Configuration of block size.

$b_1$	$b_2$	$b_3$	$b_4$	$b_5$	$b_6$	$b_7$	$b_8$
10 <sup>2</sup>	$2  imes 10^2$	$5  imes 10^2$	$8  imes 10^2$	10 <sup>3</sup>	$1.5  imes 10^3$	$2  imes 10^3$	$5  imes 10^3$

The dataset is denoted as  $d_i b_j$ . Combined with different data amounts and block sizes,  $d_i b_j$  has different block depths. The block depth refers to the total amount of blocks within the blockchain.

## 6.3. Experiment Setup

To assess the impact of the dual-layer index on both construction and query performance, we conducted four series of experiments. Each series consisted of two parts: evaluating the dual-layer index and comparing the dual-layer index with the non-layer index implementation, which was the native Bitcoin implementation.

## (1) Block Construction Cost Experiment

We compared the block construction cost of the dual-layer index with the native Bitcoin block construction cost in two aspects: construction time and block storage usage. The experiment was divided into two groups. The first group consisted of  $d_ib_j i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$  and  $j \in \{3, 5, 7\}$  to verify the influence of the data amount on the block construction. The second group consisted of  $d_ib_j i \in \{3, 5, 7\}$  and  $j \in \{1, 2, 3, 4, 5, 6, 7, 8\}$  to verify the impact of the block size on the block construction. (2) Data Amount Influence Experiment

We aimed to evaluate the influence of the data amount on traceability query efficiency. In the beginning, we focused on the dual-layer index structure only. The datasets used were  $d_ib_j$ , where  $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$  and  $j \in \{3, 5, 7\}$ . Then, we compared the traceability query efficiency between the dual-layer index structure and the non-layer index structure. The datasets used were  $d_ib_5$ , where  $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$ . For all these cases, the query execution time was recorded for result analysis. In this experiment, the traceability query consisted of 1000 transactions. In each dataset, these data were randomly distributed.

### (3) Block Depth Influence Experiment

According to the equation  $DataAmount = BlockSize \times BlockDepth$ , we used the block depth to represent  $d_ib_j$ . We recorded the block depth as  $p_i$ . The datasets used for this experiment are shown in Table 3. Then, we recorded the number of transactions for the traceability querying, namely, the query case, as  $q_j$ . The datasets used for this experiment are shown in Table 4.

In this experiment, we aimed to analyze and compare the influence of block depth on query efficiency. To increase the depth of the block, we quantified the data to be queried and then added non-queried data to create a new dataset. Specifically, we utilized the dataset  $p_iq_j$ , where  $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$  and  $j \in \{1, 2, 3\}$ , to evaluate and analyze the traceability query based on the block index method. Additionally, we selected  $p_i$ , where  $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$ ; and  $q_j$ , where  $j \in \{2\}$ , to verify the traceability query based on the dual-layer index and non-layer index methods.

Table 3. Configuration of block depth.

$p_1$	$p_2$	$p_3$	$p_4$	$p_5$	$p_6$	$p_7$	$p_8$
10 <sup>1</sup>	$1  imes 10^2$	$2  imes 10^2$	$5  imes 10^2$	10 <sup>3</sup>	$2  imes 10^3$	$5  imes 10^3$	$5 imes 10^4$

Table 4. Configuration of query case.

<i>q</i> <sub>1</sub>	<i>q</i> <sub>2</sub>	<i>q</i> 3
$5 \times 10^2$	$1  imes 10^3$	$2  imes 10^3$

#### (4) Block Size Influence Experiment

In this experiment, we compared and analyzed the effect of block size on query efficiency. Specifically, we utilized the dataset  $d_ib_j$ , where  $i \in \{3, 5, 7\}$  and  $j \in \{1, 2, 3, 4, 5, 6, 7, 8\}$ , to evaluate and analyze the traceability query based on the block index method. Additionally, we utilized the dataset  $d_ib_j$ , where  $i \in \{5\}$  and  $j \in \{1, 2, 3, 4, 5, 6, 7, 8\}$ , to verify the traceability query based on the dual-layer index and non-layer index methods. We then analyzed and compared the execution time of these methods. The distribution of the traceability data was similar to that in (2).

### (5) Extreme Case Experiment

In this experiment, the aim was to evaluate the effectiveness of the dual-layer index structure with a large data amount. Thus, the large data amount was the influential factor, denoted as  $D_i$ . The set of large data used in this experiment is listed in Table 5.

**Table 5.** Configuration of large amounts of data.

<i>D</i> <sub>1</sub>	<i>D</i> <sub>2</sub>	$D_3$	$D_4$
$2  imes 10^7$	$5  imes 10^7$	10 <sup>8</sup>	$2 imes 10^8$

#### (6) Complex Scenario Experiment

In real-world supply chain applications, different suppliers may provide raw materials with varying prices and qualities, which are mixed in batches to produce final products. During this process, there exist disassembly and assembly operations, which further complicate the traceability of the supply chain [38–40]. In the food supply chain, traceability entails the capacity to monitor and trace food, feed, food-producing animals, or substances intended for use in food or feed throughout all stages of production, processing, and distribution. As the supply chain involves several intermediaries between the initial grower of the food material and the end consumer, it is vital to document information on the movement of food items through intermediate processes.

During the intermediate processes, there can be a one-to-many or many-to-one relationship. This phenomenon is denoted as transaction bifurcation, where products are synthesized or decomposed. To address this issue at the application level, the corresponding upstream product source should be added to the product attribute. However, if the product attribute is restricted, a solution is sought at the index structure level. Namely, we need to modify the content of TraceID in the dual-layer index structure.

In our experiment, we investigated the one-to-many and many-to-one relationships between the upstream and downstream products. For the one-to-many relationship, we added extra suffix identifiers to the TraceID to differentiate between different downstream products. For many-to-one relationships, we included extra separators within the TraceID to distinguish between various upstream product sources. To account for the impact of transaction bifurcation, we denoted it as an influencing factor ( $F_i$ ). The amount of bifurcation shows the number of upstream products involved in synthesis or the number of downstream products generated from decomposition. Table 6 presents the bifurcated dataset employed in our experiment. Additionally, we specified a fixed traceability length of 10, which could reflect the number of intermediaries involved in the bifurcation phenomenon.

Table 6. Configuration of bifurcation amount.

F <sub>1</sub>	<i>F</i> <sub>2</sub>	F <sub>3</sub>	F <sub>3</sub>
1	2	3	4

## 6.4. Experimental Results and Analysis

This section provides details of our experimental results and analysis. In the experiments, we imported the generated dataset into two simulation systems: one based on the native blockchain architecture and the other based on the dual-layer index structure. In these simulation systems, blocks were stored as individual files on the disk, corresponding to a single file. The construction process involved writing the relevant datasets into the simulation systems and building the blocks accordingly. The traceability query process was performed by traversing all data or querying traceability objects based on the index. To clearly illustrate the experimental result, we introduced three indicator functions:

- *C*(*L*, *M*): block construction time on dataset *L* with index type *M*. While importing the dataset into the simulation system, we calculated the block generation time by obtaining the operation system's timestamp.
- S(L, M): storage of the simulation system on dataset L with index type M. Once the data set was imported into the simulation system, we calculated the block storage occupancy by obtaining the storage size of the block files in the simulation system.
- *T*(*L*, *M*): execution time of traceability queries on dataset *L* with index type *M*. After
  importing the dataset into the simulation system, we conducted traceability queries
  on the simulation system and calculated the time required for traceability queries by
  obtaining the operation system's timestamp.

In the above indicators,  $\mathcal{M} \in {\mathcal{D}, \mathcal{N}}$ , in which  $\mathcal{D}$  indicates the dual-layer index, and  $\mathcal{N}$  indicates the non-layer index, namely the original blockchain structure. The dataset  $\mathcal{L}$  refers to  $d_i b_j$  or  $p_i q_j$ .

## 6.4.1. Index Construction Cost Experiment

### (1) Construction Time Result

The results regarding the block construction time with the dual-layer index are presented in Figure 3a,b. It can be observed that  $C(d_ib_j, \mathcal{M})$  significantly increases with increasing *i* but gradually decreases with increasing *j*, with a diminishing rate of change. Therefore, we concluded that for a fixed block size, the data amount has a substantial impact on the block construction time. An increase in data amount leads to a longer block construction time. This is because the dual-layer index introduces time overhead, including maintaining temporary traceability tables and adding internal index fields. On the other hand, the block size has a weaker influence on construction time. When the data amount is constant, as block size increases, block construction time decreases. Because the data field of the block header is fixed and does not change with the block size, the larger the block size, the smaller the number of blocks under the same data amount, and the smaller the fixed block header. The smaller fixed block header reduces the time consumption, whereas a larger fixed block header increases the construction time consumption.



**Figure 3.** Experimental result regarding block construction time with dual-layer index under (**a**) different data amounts and (**b**) different block sizes; construction time comparison of the dual-layer index and non-layer index under (**c**) different data amounts and (**d**) different block sizes.

A comparison of the construction time between the dual-layer index and non-layer index is shown in Figure 3c,d. Normally,  $C(d_ib_j, D)$  is about 2.7 times larger than  $C(d_ib_j, M)$ . Thus, we concluded that constructing a block with the dual-layer index takes approximately 2.7 times longer than that of a non-layer index block when the data size is kept constant.

This is because the MB+ tree within the dual-layer index block needs to acquire data from the temporary traceability table and update it during the construction process.

## (2) Block Storage Result

The results for block storage usage are presented in Figure 4a,b. It can be observed that  $S(d_ib_j, \mathcal{M})$  significantly increases as *i* increases. On the other hand,  $S(d_ib_j, \mathcal{M})$  remains constant as *j* changes. Therefore, we concluded that, for a fixed block size, the data amount has a substantial impact on block storage use. This is because the dual-layer index introduces storage overhead, including the temporary traceability table and internal index fields. When the data amount is constant, the block size has little effect on the storage space needed after building the block. This is the influence of the fixed block header fields. The smaller the block, the larger the number of block headers, which increases the storage consumption. Otherwise, it reduces storage consumption.



**Figure 4.** Experimental result of block storage usage of the dual-layer index under (**a**) different data amounts and (**b**) different block sizes; storage comparison of the dual-layer index and non-layer index under (**c**) different data amounts and (**d**) different block sizes.

The comparison of block storage usage between the dual-layer index and non-layer index is shown in Figure 4c,d. It can be observed that  $S(d_ib_j, D)$  is slightly larger than  $S(d_ib_j, N)$ . Hence, we drew the conclusion that blocks with the dual-layer index take up slightly more storage space than blocks without the index for the same amount of data. The reason for this is that the root nodes and intermediate nodes of the MB+ tree in the dual-layer index block add pointer and keyword information based on the non-layer index

tree. Additionally, the leaf nodes of the dual-layer index block add an external index and maintain a temporary traceability table, resulting in more storage space usage.

### 6.4.2. Data Amount Influence Experiment

In this experiment, T-Dual-Layer  $(d_i b_j)$  was used for querying based on the block index and T-non-layer  $(d_i b_j)$  was used for query based on the non-layer index under the  $d_i b_j$  test data set.

Figure 5a shows the traceability query time of the dual-layer index ( $T(d_ib_j, D)$ ), where  $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$  and  $j \in \{3, 5, 7\}$ . It is observed that with the increase in i,  $T(d_ib_j, D)$  also increases. This indicates that for a fixed block size, the larger the amount of data, the more time required for a traceability query based on the dual-layer index. This is because the larger the data amount under a fixed block size, the larger the number of blocks. The query case is more likely to be distributed across multiple blocks, resulting in more time-consuming jumps through external indices.

Figure 5b shows a comparison of the traceability query time between the non-layer index( $T(d_ib_j, \mathcal{N})$ ) and dual-layer index, where  $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$  and j = 5. We can see that  $T(d_ib_j, \mathcal{D})$  is always smaller than ( $T(d_ib_j, \mathcal{N})$ ). When j = 5,  $T(d_ib_j, \mathcal{D}) \approx 7.5 \times T(d_ib_j, \mathcal{N})$ . It shows that the efficiency of traceability queries based on the dual-layer index is much higher than that of the non-layer index.



**Figure 5.** Experimental result of the influence of data size. (**a**) query time of dual-layer index under different data amounts; (**b**) query time comparison of dual-layer index and non-layer index under different data amounts.

## 6.4.3. Block Depth Influence Experiment

Figure 6 shows the traceability query time of the dual-layer index  $(T(p_iq_j, D))$ , where  $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$  and  $j \in \{1, 2, 3\}$ . It can be observed that with the increase in i,  $T(p_iq_j, D)$  is unchanged under different j values. This indicates that, for a fixed query case, the traceability query time is independent of the block depth based on the dual-layer index. This is because the external index records the latest trace object. No matter how much data are appended, the index query directly starts from the last place where the trace object appears. It can also be observed that  $T(p_iq_j, D)$  increases with increasing j. This indicates that the more data in the query case, the more time the traceability query takes. This is because, at the same block depth, more data in the query case lead to a longer query process.



Figure 6. Query Time of the dual-layer index with different block depths.

Table 7 compares the traceability query execution time between the non-layer index and dual-layer index for various block depths. From the table, it is evident that, except for the first case (BlockDepth = 10), the execution time for the dual-layer index is consistently lower than that for the non-layer index for the same block depth. This is because as the block depth increases, the blockchain becomes longer. However, the dual-layer index avoids a one-by-one block query through the block jump process. Thus, increasing the length of the blockchain does not affect the traceability query time of the dual-layer index for specific target data queries. On the other hand, because queries on the non-layer index require a block-by-block comparison, the traceability query efficiency decreases as the length of the blockchain increases.

Table 7. Query time comparison of dual-layer index and non-layer index for different block depths.

Block Depth	10	100	200	500	1000	2000	5000	10,000
non-layer Index (s)	0.686	6.04	11.975	30.003	58.634	115.321	284.665	658.633
Dual-Layer Index (s)	0.998	0.992	0.99	1.001	0.995	0.993	0.994	0.997

### 6.4.4. Block Size Influence Experiment

Figure 7a illustrates the effect of block size on the traceability query time of the duallayer index. As the results show, as *j* increases,  $T(d_ib_j, D)$  also proportionally increases, indicating that larger block sizes result in longer traceability query times for the dual-layer index. This is because larger block sizes increase the number of jumps between blocks, which slows the query.

Figure 7b compares the traceability query time between the non-layer index and the dual-layer index for different block size configurations. Notably, the traceability query time  $T(d_ib_j, D)$  is consistently lower than  $T(d_ib_j, N)$ . Moreover, when *i* is kept constant,  $T(d_ib_j, N)$  decreases as *j* increases while  $T(d_ib_j, D)$  increases. This observation suggests that the traceability query efficiency of the non-layer index decreases for larger block sizes for a given amount of data. This is because the dual-layer index structure primarily optimizes the block jump process. As the data amount increases, the number of blocks decreases, and the time required for block jumps in the traceability query process of the non-layer index structure decreases, thereby reducing the optimization effect of the dual-layer index on traceability queries.



**Figure 7.** Experimental result of the influence of block size. (a) Query time of dual-layer index for different block sizes; (b) Query time comparison of the dual-layer index and non-layer index for different block sizes.

## 6.4.5. Extreme Case Experiment

Table 8 presents the query time, construction time, and storage occupation of the duallayer index structure for various large datasets. Our experimental findings suggest that even with large data amounts, the query time, construction time, and storage occupation of the dual-layer index structure exhibit linear variation. This indicates that the duallayer index structure can efficiently process large data amounts, as demonstrated by our experimental results.

**Table 8.** Query time, construction time, and construct storage of dual-layer index with different data amounts.

Data Amount	$2 imes 10^7$	$5 imes 10^7$	10 <sup>8</sup>	$2 imes 10^8$
Query Time (s)	165.816	423.128	816.563	1655.432
Construction Time (s)	1579.213	3910.875	7936.125	17,136.263
Construct Storage (GB)	6.58	14.13	29.16	61.59

6.4.6. Complex Scenario Experiment

Table 9 presents the execution time of traceability query under complex scenarios such as the one-to-many scenario and the many-to-one scenario.

Table 9. Query time comparison of one-to-many and many-to-one for different bifurcation factors.

<b>Bifurcation Factor</b>	1	2	3	4	5
One-To-Many	0.124	0.118	0.102	0.136	0.122
Many-To-One	0.135	1.545	42.598	469.312	3289.885

In the one-to-many scenario, the query time remains constant for different bifurcation factors. This finding reveals that the traceability time in a one-to-many scenario is independent of the bifurcation factor when the traceability length is constant. This can be attributed to the fact that in a one-to-many relationship, the initial product can be perceived as gradually decomposing into subproducts. Despite the number of subproducts generated, the traceability from any subproduct always returns to the same length and, ultimately, back to the source product, thereby keeping the query time unchanged.

In the many-to-one scenario, the query time increases with the increase in the bifurcation factor. This finding indicates that the traceability time in the many-to-one scenario positively correlates with the bifurcation factor when the traceability length is constant. In a many-to-one relationship, the product is perceived as gradually merging with multiple raw materials. The presence of multiple raw materials requires searching for upstream raw materials based on the final merged product, which necessitates multiple backtracking. The larger the number of bifurcation factors, the more raw materials are involved, and the more backtracking is required, leading to an increase in the query time.

In practical production activities, it is common to find one-to-many and many-toone relationships in supply chains. However, mixed queries are local traceability queries involving one-to-many and many-to-one relationships. Thus, the fundamental principles remain unchanged.

## 7. Discussion

This section starts with a summary of our experimental results. Then, we illustrate the implementation scenarios of the dual-layer index structure in a practical environment. Furthermore, we compare our work with other blockchain-based supply chain traceability research. In the end, we discuss the assumptions and limitations of our method.

## 7.1. Discussion of Experimental Results

The construction of a dual-layer index structure incurs additional time and space overhead. Based on the experimental results presented in Section 6.4.1, we concluded that the 2.5 to 3.4 times more time overhead is required for constructing a dual-layer index structure compared with the original blockchain structure. However, the gap in time overhead narrows as the block size increases. Similarly, the space overhead is 1.1 to 1.2 times greater than that of the original blockchain structure, and the gap decreases as the block size increases. Though there is overhead, constructing a dual-layer index structure can significantly enhance the efficiency of traceability queries. As seen from the experimental results in Sections 6.4.2 and 6.4.4, the traceability query efficiency under the dual-layer index structure is 7.1 to 7.9 times higher than that of the original blockchain when the traceability data account for a certain proportion (set at 10% in this experiment) of the total data. The gap in query efficiency increases as the block size decreases. In contrast, the experimental results in Section 6.4.3 show that the query efficiency under the dual-layer index structure is determined by the amount and distribution of traceability data rather than the total amount of data. The experimental results in Section 6.4.5 indicate that the dual-layer index structure can efficiently process large amounts of data, as demonstrated by the linear variation observed in the query time and storage occupation for varying dataset sizes. Section 6.4.6 shows the query time is independent of the bifurcation factor in the one-to-many scenario. In contrast, the query time increases with the increase in the bifurcation factor in the many-to-one scenario. In a one-to-many scenario, the traceability from any subproduct always returns to the same length and ultimately back to the source product; in a many-to-one scenario, the presence of multiple raw materials requires multiple backtracking, which can be interpreted as multiple traceability queries.

In general, the dual-layer index structure changes the process of obtaining traceability data by replacing block traversal in the original blockchain with external block jumps based on the external index and fast queries within blocks based on the internal index. The dual-layer index reduces query time and improves query efficiency for traceability data. Although constructing the index incurs some time overhead and requires slightly more storage space, the benefits of improving query efficiency for traceability data far exceed these costs.

#### 7.2. Implementation Discussion

The internal index requires modifications of the block structure of blockchains, which makes it impractical to directly implement the dual-layer index in practical applications. Two scenarios for the implementation of the dual-layer index structure are proposed.

- For a novel food supply chain, the initial step is to modify the MB+ tree block structure. This alteration is imperative for storing the internal index and a part of the external index. Following this, the location for storing the external index must be determined. Three main options exist: (1) If a blockchain system supports a state database, such as Fabric, Ethereum, or Partiy, the external index can be stored within this database.
   (2) Another option is to make use of a special block that is designated for external index storage, but this may lead to query time overhead. (3) Alternatively, an external database can be introduced to store the external index, though this approach may introduce security and tampering problems. In the end, to ensure the maintenance of the system, an intelligent contract must be created that writes the traceability data and effectively maintains the dual-layer index structure.
- For an existing food supply chain, the first step is to create a novel blockchain environment that supports the dual-layer index feature. Subsequently, the traceability data should be exported from the original supply chain system and imported into the novel blockchain system. However, this process may entail a cross-chain operation. If the underlying blockchain systems are identical, a data transformation intelligent contract would be necessary. Otherwise, the traceability data have to be exported to local storage and imported to the novel blockchain system later. However, the use of local storage for data exporting poses potential issues. The absence of adequate data protection mechanisms may result in tampering problems.

To illustrate the implementation details, we introduced a case study based on the European food supply chain, a blockchain system for ensuring traceability, as described in [41]. The system allows consumers to reconstruct the entire history of a product up to its origin to verify its health and quality through a simple QR code scan. The general agricultural food supply chain consists of all activities at different stages, namely, production, processing, distribution, and consumption. The proposed solution considers the continuity of different stages and events in the supply chain. Following the four stages mentioned above, the case study was a rice supply chain.

(1) In the production stage, farmers add crop information to the blockchain, including the seed source, production site, fertilizer use, harvest time, product batch, etc. At this stage, a TraceID is defined, which can be defined based on the production site and production batch. Rice produced in the same location and batch is assigned the same TraceID.

(2) In the processing stage, processors first store transaction records with farmers on the blockchain, including information on both the parties and the traded items. Then, processors need to store information about the rice in the drying, cleaning, and storage processes, including the product batch, use of chemicals, impurity content, removal rate, inventory number, product source, storage time, quality inspection report, product category, product quantity, etc. In the processing stage, rice may undergo branching or merging due to different storage and processing batches. For example, rice with different TraceIDs may be stored under the same conditions, or rice with the same TraceID may undergo different processing batches. In this case, the TraceID of the branching product can be identified by adding an additional identifier. The TraceID of the merged product can be separated by a delimiter, indicating that the merged result comes from rice with multiple TraceIDs.

(3) In the distribution stage, distributors first store transaction records with processors on the blockchain, including information on both parties and the traded items. Distributors then record transportation details, tracking information, storage conditions (such as temperature and humidity), and transportation time. All transactions between distributors and retailers, as well as all relevant information, are written into the blockchain platform. In this stage, packaged rice may also undergo branching or merging due to the different transportation processes. The handling of the TraceID at this stage is the same as in the processing stage.

(4) In the consumption stage, detailed information about each bag of rice is recorded, including its current quality, expiration date, and storage conditions. The TraceID from the distribution process is used in this stage and is recorded on the bar code attached to the

final product packaging. Customers who purchase the rice can scan the bar code to obtain the TraceID and trace the origin of the rice.

To analyze this case using a Hyperledger-Fabric-based dual-layer indexing structure, we needed to make technical modifications to the Hyperledger Fabric blockchain. Based on the characteristics of Hyperledger Fabric, the main modifications can be divided into three parts: modifications of the block structure, modifications of interface implementations, and additional interface implementations.

(1) Modifications of the block structure: The block structure in Hyperledger Fabric consists of a block header and a block body. The block header does not require any modifications, and the dual-layer indexing structure only needs to modify the storage structure of the transaction data in the block body to an MB+ tree structure. External indices are introduced into the stored transaction data in the block body through two new fields: TraceID and *preTraceHash*. Because the Hyperledger Fabric chaincode Query System Chaincode (QSCC) comes with the *GetBlockByHash* interface, block data can be directly retrieved by the block hash. Therefore, the *preTraceHash* can assist block-to-block jumping, reducing the modification cost. (a) Users provide the TraceID that is recorded in the QR code or bar code of the product. The TraceID is stored in the system as product information is recorded. For example, in the rice supply chain, farmers in the production stage, processors in the processing stage, distributors in the distribution stage, and retailers in the consumption stage need to provide the TraceID when recording intermediate product information on the blockchain. (b) *preTraceHash* is updated with the temporary traceability table. Hyperledger Fabric supports a state database, so the table in the dual-layer index structure is stored in the state database. The table contains information such as TraceID, TransactionID, and *preTraceHash*. When a transaction with a corresponding TraceID is first inserted, the *preTraceHash* is null. During the Fabric transaction process, the orderer nodes generate block sorting, and the peer node synchronizes the block data to the state database. Therefore, the source code of the synchronization process needed to be modified to synchronize TraceID, TransactionID, and *preTraceHash* to the state database during this process.

(2) Modifications of interfaces: Modifications of the block structure affect the interfaces related to block operations. In Hyperledger Fabric, some interfaces are based on the state database to obtain data, such as *PutState* and *GetState*. Because the storage format of the state database remains unchanged, these interfaces do not need to be modified. However, some interfaces are based on block queries in the QSCC, such as *GetBlockByNumber*, *GetTransactionById*, *GetBlockByHash*, etc. Due to the modifications in the block structure, the interface implementation of QSCC needed to be modified.

(3) Additional interface implementations: To improve the efficiency of traceability queries of the dual-layer index structure, an additional traceability query interface needed to be implemented. The traceability query interface was designed to query the complete history of a product based on its TraceID. The query process involves two parts: (a) The first part involves finding the block with the last TraceID based on the temporary traceability table in the state database, parsing the block data, and searching transactions with the same TraceID using the MB+ tree query. (b) The second part involves jumping to the corresponding block for the query based on the *preTraceHash*. This step is repeated until the *preTraceHash* field of the transactions in the result set is null. To integrate these two steps into a query interface, the first step can be implemented using the Fabric native interface *GetState*, as the temporary traceability table is stored in the state database. The second step can be implemented using the *GetBlockByHash* interface in QSCC, which retrieves the corresponding block data and parses the transaction set. By implementing these two steps in a query interface, users can easily query the complete history of a product based on its TraceID.

Combining the Fabric transformation based on the dual-layer index structure, we calculated the transformation cost. Regarding the cost of modifying the code, the dual-layer index structure involves modifying the block body structure; it is necessary to insert the

TraceID, TransactionID, and *preTraceHash* fields of the temporary traceability table into the synchronous block data. However, in Fabric, changing the block body structure affects the code modification of the block, including block data storage and query. Therefore, the following source code needs to be modified: the block body structure, block storage format, block data synchronization, etc. The modification involves the following packages: common package, core package, and peer package, resulting in a large range of modifications. Fabric is a mature blockchain technology with a complete identity authentication mechanism, encryption mechanism, and consensus algorithm. These mechanisms are unrelated to the block structure, and the dual-layer index structure does not involve modifying the above mechanisms. Therefore, the code modification introduced by the dual-layer index structure accounts for a relatively small amount of Fabric source code.

Regarding the maintenance cost, after the block structure transformation, the data stored in the block follow the standard transaction insertion logic without additional user operation. However, the temporary traceability table must be maintained in realtime. Specifically, the TraceID of the temporary traceability table needs to be updated whenever the transaction is inserted. Regarding storage cost, in our study case, the native Fabric occupies 36.4 GB, while the Fabric with the dual-layer index occupies 45.9 GB, approximately 1.26 times larger the original block storage cost. This result is consistent with the general experimental results. Regarding query performance improvement, the original Fabric stores case data and performs traceability queries. It takes 1534.16 s to analyze data by traversing blocks, whereas it takes only 375.53 s to use the additional traceability query interface for traceability queries under the dual-layer index structure. However, in actual cases, traceability queries require network communication between clients and the blockchain, which consumes extra time due to network I/O compared with the experimental results.

### 7.3. Comparison with Related Methods

We compared the dual-layer index structure with related methods based on method type, security, sensitive data protection, storage overhead, and traceability query influencing factors. The comparison results are summarized in Table 10.

Ref	Research Object	Method Type	Security	Sensitive Data Protection	Storage Overhead	Traceability Query Influencing Factors
Ours	Blockchain Traceability Query	Index	Blockchain	No	Low	Traceability Length
[42]	Blockchain Traceability Query	Index	Blockchain	No	Low	Data Amount
[43]	Blockchain Traceability Query	Index	Blockchain	No	Low	Data Amount
[26]	Blockchain Query	External Database	External Database	No	High	Data Amount
[14]	Blockchain Query	External Database	External Database	No	High	Data Amount
[44]	Application of Supply Chain	Application	Blockchain	Yes	Low	-
[45]	Application of Supply Chain	Application	Blockchain	Yes	Low	-

Table 10. Comparison results.

In addition to our method, those of Lai et al. [42] and Qin et al. [43] use an index structure to optimize traceability query efficiency. Lai et al. [42] designed an index structure independent of the Merkle tree to support efficient intrablock queries and create new fields in the block header to optimize the interblock query. However, the method of Lai et al. still needs to traverse all the blocks to compare the added block header fields. The method of Qin et al. [43] also needs to find the first traceability object by traversing the block. Moreover, their experiments showed that the efficiency of the traceability query is related to the total amount of data under their indexing methods. The advantage of our method is that traceability time is only affected by traceability length.

Li et al. [26] and Peng et al. [14] achieved the advancement of additional query middleware in the blockchain, but both studies introduced external database storage, which may cause security problems.

Martintoni et al. [44] proposed a sensitive data protection strategy based on an authentication mechanism in an alliance chain application. Meanwhile, a more comprehensive sensitive data management approach was developed by Cha et al. [45], which adopts a key escrow encryption system for data protection. Notably, the dual-layer index structure only makes modifications to the blockchain structure to boost the performance of the traceability query, thereby preserving the inherent features of the blockchain. Therefore, we did not consider the sensitive data projection in the dual-layer index structure. The aforementioned data protection strategies are complementary to our method.

### 7.4. Assumptions and Limitations

After analyzing the characteristics of the dual-layer index structure and reviewing the experimental results, we identified several assumptions and limitations of this approach.

## (1) Assumptions

Firstly, the implementation of the dual-layer index structure requires a transformation of the blockchain to accommodate changes in the block structure. This transformation can be directly upgraded for private and alliance chains; however, for public chains, the requisite technical transformation authority must be obtained.

Secondly, it is recommended to have technical support similar to the state database, which is readily available in many existing blockchain technologies such as Fabric and Ethereum. By using a state database, the external index can be stored in it for maintenance, maximizing the benefits of the dual-layer index structure. However, if a stateless database is used, there might be a need to store the external index in the block, which could result in a performance decrease.

Thirdly, the benefits of the dual-layer index structure are particularly evident when there is a significant user base. It is easy to distribute traceability data across nonadjacent blocks and enhance the efficiency of queries.

Fourthly, it is essential to determine an appropriate block size. When the total data amount is fixed, a smaller block size leads to a larger number of blocks. Because each block contains an MB+ tree structure (internal index), this results in increased overhead storage. Conversely, a larger block size leads to a higher MB+ tree and reduces the number of blocks needed. However, the effectiveness of the dual-layer index structure is weakened due to longer intrablock searches and fewer block jumps during traceability queries. Therefore, it is necessary to make a trade-off decision.

#### (2) Limitations

The security and tampering issues of the dual-layer index structure are dependent on the security features of the underlying blockchain technology. The internal index MB+ tree retains the properties of the Merkle tree, allowing it to expose data tampering in a timely manner. Likewise, the temporary traceability tables rely on the storage of the blockchain itself. Therefore, the dual-layer index structure does not introduce any additional security risks to the blockchain, and its security is entirely dependent on the reliability of the underlying blockchain technology.

One potential issue with the dual-layer index structure is the lack of protection mechanisms for data privacy, especially sensitive data. The data involved in the supply chain include not only product traceability information but also private data that only related enterprises can access, such as transaction information. For competitive enterprises, data privacy is a crucial issue. The external index uses the self-increasing TraceID as the index's primary key. This may lead to the leakage of sensitive data, as tracing all products' specific traceability paths can be easily achieved at a low cost (by traversing the external index), which could expose the cooperative relationships between companies. One solution to this issue is to introduce other TraceID-encoding schemes or encryption methods to encrypt the TraceID. The TraceID could then be accessed through specific decoding means when tracing the product's traceability path, which can effectively protect sensitive data and ensure that only authorized parties can access traceability information.

Under extreme traceability scenarios, the storage cost can sharply increase. When the total amount of data is fixed, an increase in traceable data leads to an increase in external index storage costs. Additionally, when there are too many traceability branches, the TraceID in the external index may contain a large number of repeated prefixes, exacerbating external index storage costs. In the case of one-to-many traceability, a snowflake-like pattern may emerge, further increasing storage costs. For many-to-one traceability, using separators to identify upstream products can make the internal index invalid. To address these challenges, a new external TraceID index coding method could be introduced. At the application level, the TraceID of upstream products can be recorded as part of product attributes. This can help to reduce storage costs and ensure the external index remains accurate and efficient, even in extreme traceability scenarios.

#### 8. Conclusions

In recent years, food quality and security issues have become increasingly serious. Ensuring transparency and traceability in the food supply chain is crucial for maintaining food safety, quality control, compliance, and consumer trust. Compared with traditional centralized management approaches, blockchain technology offers a better solution with its decentralized and traceable characteristics.

To this end, we proposed a dual-layer index based on blockchain that can be applied to supply chain traceability systems, providing an efficient solution to support traceability. We designed an MB+ tree data structure to improve the internal block index and used a temporary traceability table as the external block index. Our experiments demonstrated that the traceability query capability of the dual-layer index is significantly improved, with an average increase in traceability query efficiency of seven to eight times. Furthermore, we verified that the dual-layer index structure can function generally under large datasets and complex scenarios. The results of these experiments demonstrated that the query efficiency of the dual-layer index structure in complex scenarios still adheres to general patterns and is only related to traceability length.

To effectively implement the dual-layer index structure for supply chain traceability, several assumptions must be met, including transformation conditions corresponding to blockchain technology, support for the state database, a sufficient number of users, and reasonable block size parameters. Meeting these assumptions is essential to maximize the benefits of the dual-layer index structure while minimizing overhead costs and ensuring efficient storage and retrieval of traceability data. The dual-layer index structure also has some limitations, which need to be addressed. The security of the structure entirely depends on the underlying blockchain technology, and sensitive data may be exposed if not protected. The temporary traceability table may reveal the cooperative relationships between companies, but this can be addressed by encoding or encrypting the TraceID. In extreme traceability scenarios, the structure may introduce significant storage overhead, requiring storage data or modifying the TraceID-encoding method at the application level.

In future work, we aim to improve the dual-layer index structure by addressing its existing limitations. These improvements include enhancing the protection of sensitive data by incorporating existing sensitive-data-protection methods, developing a new TraceID encoding method combined with cryptography techniques, and investigating new storage structures to improve its scalability while maintaining query efficiency. These enhancements aim to enhance the security, scalability, and efficiency of the dual-layer index structure, making it more applicable for practical use. Furthermore, we plan to explore integrating index structures with product attributes in the future, which may involve incorporating new and emerging data formats. Such integration could help minimize the storage overhead of additional indices currently in use.

**Author Contributions:** Conceptualization, C.G., Y.L., M.N. and J.S.; data collection, C.G. and Y.L.; formal analysis, C.G., Y.L., M.N. and J.S.; investigation, Y.L. and M.N.; methodology, C.G. and Y.L.; project administration, C.G.; resources, C.G. and Y.L.; validation, M.N.; visualization, Y.L. and M.N.; writing—original draft, C.G. and Y.L.; writing—review and editing, C.G. and J.S. All authors have read and agreed to the published version of the manuscript.

**Funding:** This paper is supported by the Fundamental Research Funds for the Central University (CN) N2317005.

Data Availability Statement: Data are contained within the article.

**Conflicts of Interest:** The authors declare no conflict of interest.

## References

- 1. Liu, P.; Zhang, Z.; Li, Y. Investment Decision of Blockchain-Based Traceability Service Input for a Competitive Agri-Food Supply Chain. *Foods* **2022**, *11*, 2981. [CrossRef]
- 2. Latino, M.E.; Corallo, A.; Menegoli, M.; Nuzzo, B. An integrative conceptual framework of food certifications: Systematic review, research agenda, and macromarketing implications. *J. Macromark.* **2022**, *42*, 71–99. [CrossRef]
- Khanna, A.; Jain, S.; Burgio, A.; Bolshev, V.; Panchenko, V. Blockchain-Enabled Supply Chain platform for Indian Dairy Industry: Safety and Traceability. *Foods* 2022, 11, 2716. [CrossRef] [PubMed]
- 4. Kechagias, E.P.; Gayialis, S.P.; Papadopoulos, G.A.; Papoutsis, G. An Ethereum-Based Distributed Application for Enhancing Food Supply Chain Traceability. *Foods* **2023**, *12*, 1220. [CrossRef]
- 5. Islam, S.; Cullen, J.M. Food traceability: A generic theoretical framework. Food Control 2021, 123, 107848. [CrossRef]
- 6. Wang, L.; Xu, L.; Zheng, Z.; Liu, S.; Li, X.; Cao, L.; Li, J.; Sun, C. Smart Contract-Based Agricultural Food Supply Chain Traceability. *IEEE Access* 2021, 9, 9296–9307. [CrossRef]
- Tian, F. An agri-food supply chain traceability system for China based on RFID & blockchain technology. In Proceedings of the International Conference on Service Systems and Service Management, Kunming, China, 24–26 June 2016; IEEE: Kunming, China, 2016; pp. 1–6. [CrossRef]
- 8. Yiu, N.C.-K. Toward Blockchain-Enabled Supply Chain Anti-Counterfeiting and Traceability. *Future Internet* 2021, 13, 86. [CrossRef]
- Zheng, Z.; Xie, S.; Dai, H.; Chen, X.; Wang, H. An Overview of Blockchain Technology: Architecture, Consensus, and Future Trends. In Proceedings of the 2017 IEEE International Congress on Big Data (BigData Congress), Honolulu, HI, USA, 25–30 June 2017; IEEE: Honolulu, HI, USA, 2017; pp. 557–564. [CrossRef]
- Gupta, H.; Hans, S.; Aggarwal, K.; Mehta, S.; Chatterjee, B.; Jayachandran, P. Efficiently processing temporal queries on hyperledger fabric. In Proceedings of the International Conference on Data Engineering, Paris, France, 16–19 April 2018; IEEE: Paris, France, 2018; pp. 1489–1494. [CrossRef]
- 11. Sunny, J.; Undralla, N.; Madhusudanan Pillai, V. Supply Chain Transparency through Blockchain-Based Traceability: An Overview with Demonstration. *Comput. Ind. Eng.* **2020**, *150*, 106895. [CrossRef]
- 12. Kim, H.M.; Laskowski, M. Toward an ontology-driven blockchain design for supply-chain provenance. *Intell. Syst. Account. Financ. Manag.* 2018, 25, 18–27. [CrossRef]
- Xie, J.; Zhu, S.; Li, B. Research on data storage model of household electrical appliances supply chain traceability system based on blockchain. In Proceedings of the International Conference on Advanced Computational Intelligence, Wanzhou, China, 14–16 May 2021; IEEE: Wanzhou, China, 2021; pp. 179–185. [CrossRef]
- Peng, Z.; Wu, H.; Xiao, B.; Guo, S. VQL: Providing Query Efficiency and Data Authenticity in Blockchain Systems. In Proceedings of the IEEE International Conference on Data Engineering Workshops, Macao, China, 8–11 April 2019; IEEE: Macao, China, 2019; pp. 1–6. [CrossRef]
- 15. Prashar, D.; Jha, N.; Jha, S.; Lee, Y.; Joshi, G.P. Blockchain-Based Traceability and Visibility for Agricultural Products: A Decentralized Way of Ensuring Food Safety in India. *Sustainability* **2020**, *12*, 3497. [CrossRef]
- 16. Damoska Sekuloska, J.; Erceg, A. Blockchain Technology toward Creating a Smart Local Food Supply Chain. *Computers* **2022**, 11, 95. [CrossRef]
- 17. Zhang, X.; Li, Y.; Peng, X.; Zhao, Z.; Han, J.; Xu, J. Information Traceability Model for the Grain and Oil Food Supply Chain Based on Trusted Identification and Trusted Blockchain. *Int. J. Environ. Res. Public Health* **2022**, *19*, 6594. [CrossRef] [PubMed]
- Kong, L.; Chen, W.; Lv, H.; Chen, Q.; Lin, G.; Huang, S.; Deng, W. BCSChain: Blockchain-based ceramic supply chain. In Proceedings of the International Conference on Blockchain and Transtworthy System, Chengdu, China, 4–5 August 2022; Svetinovic, D., Zhang, Y., Luo, X., Huang, X., Chen, X., Eds.; Springer: Chengdu, China, 2022; Volume 1679, pp. 91–104. [CrossRef]
- Hader, M.; Tchoffa, D.; Mhamedi, A.E.; Ghodous, P.; Dolgui, A.; Abouabdellah, A. Applying integrated Blockchain and Big Data technologies to improve supply chain traceability and information sharing in the textile sector. *J. Ind. Inf. Integr.* 2022, 28, 100345. [CrossRef]
- 20. Abdallah, S.; Nizamuddin, N. Blockchain-based solution for pharma supply chain industry. *Comput. Ind. Eng.* 2023, 177, 108997. [CrossRef]

- 21. Liu, H.; Yang, B.; Xiong, X.; Zhu, S.; Chen, B.; Tolba, A.; Zhang, X. A financial management platform based on the integration of blockchain and supply chain. *Sensors* 2023, 23, 1497. [CrossRef]
- Helmer, S.; Roggia, M.; Ioini, N.E.; Pahl, C. EthernityDB—Integrating Database Functionality into a Blockchain. In Proceedings of the New Trends in Databases and Information Systems, Budapest, Hungary, 2–5 September 2018; Benczúr, A., Thalheim, B., Horváth, T., Chiusano, S., Cerquitelli, T., Sidló, C., Revesz, P.Z., Eds.; Springer: Budapest, Hungary, 2018; Volume 909, pp. 37–44. [CrossRef]
- McConaghy, T.; Marques, R.; Muller, A.; De Jonghe, D.; McConaghy, T.; McMullen, G.; Henderson, R.; Bellemare, S.; Granzotto, A. Bigchaindb: A scalable blockchain database. *White Paper*. 2016. Available online: https://gamma.bigchaindb.com/whitepaper/ bigchaindb-whitepaper.pdf (accessed on 16 April 2023).
- 24. Muzammal, M.; Qu, Q.; Nasrulin, B. Renovating blockchain with distributed databases: An open source system. *Future Gener. Comput. Syst.* **2019**, *90*, 105–117. [CrossRef]
- Zhou, E.; Sun, H.; Pi, B.; Sun, J.; Yamashita, K.; Nomura, Y. Ledgerdata Refiner: A Powerful Ledger Data Query Platform for Hyperledger Fabric. In Proceedings of the International Conference on Internet of Things: Systems, Management and Security, Granada, Spain, 22–25 October 2019; IEEE: Granada, Spain, 2019; pp. 433–440. [CrossRef]
- Li, Y.; Zheng, K.; Yan, Y.; Liu, Q.; Zhou, X. EtherQL: A Query Layer for Blockchain System. In Proceedings of the International Conference on Database Systems for Advanced Applications, Suzhou, China, 27–30 March 2017; Candan, S., Chen, L., Pedersen, T.B., Chang, L., Hua, W., Eds.; Springer: Suzhou, China, 2017; Volume 10178, pp. 556–567. [CrossRef]
- Pratama, F.A.; Mutijarsa, K. Query Support for Data Processing and Analysis on Ethereum Blockchain. In Proceedings of the International Symposium on Electronics and Smart Devices, Bandung, Indonesia, 23–24 October 2018; IEEE: Bandung, India, 2018; pp. 1–5. [CrossRef]
- Ma, S.; Cao, Y.; Xiong, L. Efficient logging and querying for blockchain-based cross-site genomic dataset access audit. BMC Med. Genom. 2020, 13, 91. [CrossRef]
- Ren, Y.; Zhu, F.; Sharma, P.K.; Wang, T.; Wang, J.; Alfarraj, O.; Tolba, A. Data Query Mechanism Based on Hash Computing Power of Blockchain in Internet of Things. *Sensors* 2019, 20, 207. [CrossRef]
- Pei, Q.; Zhou, E.; Xiao, Y.; Zhang, D.; Zhao, D. An Efficient Query Scheme for Hybrid Storage Blockchains Based on Merkle Semantic Trie. In Proceedings of the International Symposium on Reliable Distributed Systems, Shanghai, China, 21–24 September 2020; IEEE: Shanghai, China, 2020; pp. 51–60. [CrossRef]
- Niu, Y.; Zhang, C.; Wei, L.; Xie, Y.; Zhang, X.; Fang, Y. An Efficient Query Scheme for Privacy-Preserving Lightweight Bitcoin Client with Intel SGX. In Proceedings of the IEEE Global Communications Conference, Waikoloa, HI, USA, 9–13 December 2019; IEEE: Waikoloa, HI, USA, 2019; pp. 1–6. [CrossRef]
- He, K.; Shi, J.; Huang, C.; Hu, X. Blockchain Based Data Integrity Verification for Cloud Storage with T-Merkle Tree. In Proceedings of the International Conference on Algorithms and Architectures for Parallel Processing, New York City, NY, USA, 2–4 October 2020; Qiu, M., Ed.; Springer: New York, NY, USA, 2020; Volume 12454, pp. 65–80. [CrossRef]
- Xu, Y.; Zhao, S.; Kong, L.; Zheng, Y.; Zhang, S.; Li, Q. ECBC: A High Performance Educational Certificate Blockchain with Efficient Query. In Proceedings of the International Colloquium on Theoretical Aspects of Computing, Hanoi, Vietnam, 23–27 October 2017; Hung, D.V., Kapur, D., Eds.; Springer: Hanoi, Vietnam, 2017; Volume 10580, pp. 288–304. [CrossRef]
- 34. Radanliev, P.; De Roure, D. New and Emerging Forms of Data and Technologies: Literature and Bibliometric Review. *Multimed. Tools Appl.* **2023**, *82*, 2887–2911. [CrossRef]
- 35. Kamilaris, A.; Fonts, A.; Prenafeta-Boldv, F.X. The rise of blockchain technology in agriculture and food supply chains. *Trends Food Sci. Technol.* **2019**, *91*, 640–652. [CrossRef]
- Ray, P.P.; Kumar, N.; Dash, D. BLWN: Blockchain-Based Lightweight Simplified Payment Verification in IoT-Assisted e-Healthcare. IEEE Syst. J. 2021, 15, 134–145. [CrossRef]
- Huang, X.; Gong, X.; Huang, X.; Zhao, L.; Gao, K. EBTree: A B-plus Tree Based Index for Ethereum Blockchain Data. In Proceedings of the Asia Service Sciences and Software Engineering Conference, Nagoya, Japan, 13–15 May 2020; ACM: New York, NY, USA, 2020; pp. 83–90. [CrossRef]
- Schmidt, B.V.; Moreno, M.S. Traceability Optimization in the Meat Supply Chain with Economic and Environmental Considerations. Comput. Ind. Eng. 2022, 169, 108271. [CrossRef]
- Saranya, P.; Maheswari, R. Proof of Transaction (PoTx) Based Traceability System for an Agriculture Supply Chain. *IEEE Access* 2023, 11, 10623–10638. [CrossRef]
- 40. Dietrich, F.; Louw, L.; Palm, D. Blockchain-Based Traceability Architecture for Mapping Object-Related Supply Chain Events. *Sensors* **2023**, 23, 1410. [CrossRef] [PubMed]
- Baralla, G.; Pinna, A.; Corrias, G. Ensure Traceability in European Food Supply Chain by Using a Blockchain System. In Proceedings of the 2019 IEEE/ACM 2nd International Workshop on Emerging Trends in Software Engineering for Blockchain (WETSEB), Montreal, QC, Canada, 27–27 May 2019; IEEE: Montreal, QC, Canada, 2019; pp. 40–47. [CrossRef]
- Lai, C.; Wang, Y. Achieving efficient and secure query in blockchain-based traceability systems. In Proceedings of the International Conference on Privacy, Security & Trust, Fredericton, NB, Canada, 22–24 August 2022; IEEE: Fredericton, NB, Canada, 2022; pp. 1–5. [CrossRef]

- Qin, H.; Si, Y.; Xue, C.; Li, X.; Jiang, Z. A query optimization method for blockchain-based traceability system. In Proceedings of the Information Technology, Networking, Electronic and Automation Control Conference, Chongqing, China, 24–26 February 2023; IEEE: Chongqing, China, 2023; Volume 6, pp. 236–242. [CrossRef]
- Martintoni, D.; Senni, V.; Marin, E.G.; Gutierrez, A.J.C. Sensitive Information Protection in Blockchain-Based Supply-Chain Management for Aerospace. In Proceedings of the 2022 IEEE International Conference on Omni-layer Intelligent Systems (COINS), Barcelona, Spain, 1–3 August 2022; IEEE: Barcelona, Spain, 2022; pp. 1–8. [CrossRef]
- 45. Cha, S.; Baek, S.; Kim, S. Blockchain based sensitive data management by using key escrow encryption system from the perspective of supply chain. *IEEE Access* 2020, *8*, 154269–154280. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.